Dear Sir,

I am grateful for the remarks of the reviewers and am sending the point wise changes made in the manuscript (as desired) as attachments. The consent form is also being sent as an attachment. I am pasting the same along also for kind consideration:

Reviewer 1:

1. "The pathophysiologic link" The possible pathophysiologic link between the Fournier's gangrene and the scar carcinoma and the likely etiologic link though not very clear has been included in the discussion.
2. "Fournier's gangrene management need not be the thrust of the article" - The point is well taken and the details of Fournier's gangrene have been omitted from the discussion as suggested.

Reviewer 2:

We are grateful for the remarks of the reviewer and an effort has been made to incorporate the desired changes in the manuscript.

1. "Needs tightening up of English" an effort has been made to improve the English with the help of an English colleague.
2. "References not in any particular order" Needful has been done and the references have been organized as suggested.
3. "Lose 500-700 words" Needful has been done.
4. "Fig. II Poor quality replace or leave out". The figure has been left out as desired.
5. "Abstract: needs rewriting and shortening". An effort has been made to reduce the size and has been rewritten for kind consideration.
6. "Introduction" rewritten as suggested.
7. Case presentation: 3 years back has been replaced with three years previously. In correct use of capital letters for x-ray, urine etc. Needful done.
8. Discussion: has been rewritten as desired.
9. Conclusion: Rewritten as suggested.

Reviewer 3:

We are grateful for the remarks of the referee and an effort has been made to make the desired changes in the manuscript as follows.
1. Case report: "The paper is too lengthy in relation with the subject" grammar and punctuation etc. It has been checked and corrected.
2. Conceptual repetition has been taken care of.

I thank you
With kind regards
Yours sincerely
chintamani